Author's response to reviews

Title: Walk-ins seeking treatment at an emergency department or general practitioner out-of-hours service: a cross-sectional comparison

Authors:

Corinne Chmiel (Corinne.Chmiel@usz.ch)
Carola A. Huber (Carola.Huber@usz.ch)
Thomas Rosemann (Thomas.Rosemann@usz.ch)
Marco Zoller (Marco.Zoller@usz.ch)
Klaus Eichler (eich@zhaw.ch)
Patrick Sidler (patrick.sidler@waid.zuerich.ch)
Oliver Senn (oliver.senn@usz.ch)

Version: 5 Date: 21 April 2011

Author's response to reviews:

Dear BMC Series Editorial Production Team

The requested major and minor changes to the format of my manuscript have been performed, besides one requested change in the tables: visible vertical lines were not removed. I thank you in advance of making these changes for me.

Best wishes

Corinne Chmiel

Dr. med. Corinne Chmiel
Research Associate
Institute of General Practice and Health Services Research University of Zurich
Pestalozzistrasse 24
CH-8091 Zürich
Switzerland
Phone: +0041 (0)44 / 255 87 11
Fax: +0041 ( 0)44 / 255 90 97
www.hausarztmedizin.uzh.ch